Author's response to reviews

**Title:** Relationship between morphological features and kinetic patterns of enhancement of the dynamic breast magnetic resonance imaging and clinico-pathological and biological factors in invasive breast cancer

**Authors:**

Oscar Fernández-Guinea (investigacion@hospitaldejove.com)
Alejandro Andicoechea (investigacion@hospitaldejove.com)
Luis O. González (investigacion@hospitaldejove.com)
Salomé González-Reyes (investigacion@hospitaldejove.com)
Antonio M. Merino (investigacion@hospitaldejove.com)
Luis C. Hernández (investigacion@hospitaldejove.com)
Alfonso López-Muñiz (investigacion@hospitaldejove.com)
Paz García-Pravia (investigacion@hospitaldejove.com)
Francisco J. Vizoso (investigacion@hospitaldejove.com)

**Version:** 4  **Date:** 28 December 2009

**Author's response to reviews:** see over
December 28th, 2009

Dr. Francisco Vizoso
Servicio de Cirugía General
Hospital de Jove
Avda. Eduardo Castro s/n
33290-Gijón, SPAIN
Fax: 34 985103564
e-mail: investigacion@hospitaldejove.com

Sabina Alam, PhD
Scientific Editor
BMC series journals

Tel: +44 (0) 20 3192 2013
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/

Dear Sabina Alam,

Please find enclosed the re-revised version of our manuscript entitled “Relationship between morphological features and kinetic patterns of enhancement of the dynamic breast magnetic resonance imaging and clinico-pathological and biological factors in invasive breast cancer” (Reference: MS: 8418175082684879). We have revised the paper in response to all of the reviewer's minor and major comments, according to which it has been something modified. We are fully thanks their interesting comments on our manuscript.

Next, I would be pleased to answer to all the specific remarks suggested by the reviewers:

**Major revisions:**

**Figure legends:** please provide a legend for figures 1 and 2. The figure legend appears after the tables

**Table 1:** all tablels in landscape must not exceed one page. We have modified the table 1 and not exceed one page

**Minors revisions:**

**Figure cropping:** It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Figures have been cropped
Grant support statement: please incorporate this information into the Acknowledgments section. The grant support statement has been incorporate into the acknowledgments section.